Author's response to reviews

Title: High Prevalence of Substance Use among High School Adolescents in Woreta Town, Northwest Ethiopia and Associated Factors: Multi-Domain Factor Analysis

Authors:

Anteneh Messele (antenehmessele33@gmail.com)
Telake Azale (telakea@yahoo.com)
Solomon Messeret (messisol@gmail.com)

Version: 5 Date: 23 July 2014

Author's response to reviews:

Dear Editors,

We would like to thank the editors for helping us improve the quality of English language and assisting us in publishing our work to the journal.

Here is the response to the editors;

We have tried to copy edit with the help of the native English language speaker.

The English language, grammar and sentence structure of the whole manuscript is edited.

We are confident that the language quality is strongly improved now.

We look forward to hearing from you!